Author's response to reviews

Title: Leisure-time versus full-day energy expenditure: a cross-sectional study of sedentarism in a Portuguese urban population

Authors:

Diane L Gal (dgal@med.up.pt)
Ana-Cristina Santos (acsantos@med.up.pt)
Henrique Barros (hbarros@med.up.pt)

Version: 3 Date: 23 January 2005

Author’s response to reviews:

Please find enclosed the article in its revised formatting. As requested the author and affiliation title was removed from the article and the authors were linked to their affiliated institution using superscript numbering. The tables were moved to be placed after the references and the two references without all author names were updated to include all author names. The full article was also reviewed for minor mistakes.